Author's response to reviews

Title: Are elderly people with co-morbidities involved adequately in medical decision making when hospitalised? A cross-sectional survey

Authors:

Anne Ekdahl (anne.ekdahl@lio.se)
Lars Andersson (lars.andersson@isv.liu.se)
Ann-Britt Wirehn (ann-britt.wirehn@lio.se)
Maria Friedrichsen (maria.friedrichsen@liu.se)

Version: 6 Date: 14 August 2011

Author's response to reviews: see over
Sweden the 14th of August, 2011

Dear editor,
Thank you very much for still considering publishing our article: MS: 1335967199517034 “Are elderly people with co-morbidities involved adequately in medical decision making when hospitalised? A cross-sectional survey”. Anne W Ekdahl, Lars Andersson, Ann-Britt Wiréhn and Maria Friedrichsen in BioMedCentral Geriatrics.
I have now made changes according to most appropriate remarks from Referee 3. Referee 1 and 2 did not have any other comments.

Kind regards

Anne Ekdahl, MD
Corresponding author
Anne.ekdahl@lio.se

Reviewer: Jan Florin
Reviewer's report:
- The aim ought to be phrased exactly the same in the abstract and in the main text. Now there is a loss of understanding about the investigation of actual degree of control in the abstract.
  Done

- The references no 24 and 25 is not present in the main text. Reference no 24 seem to be the exact same as reference no 23. No 25 is not referenced in the main text at all.
  Corrected